Author's response to reviews

Title: Acupuncture for the Treatment of Tinnitus: A Systematic Review

Authors:

Jong-In Kim (hann8400@hanmail.net)
Jun-Yong Choi (orientdoct@gmail.com)
Dong-Hyo Lee (secretop17@naver.com)
Tae-Young Choi (superyoung@gmail.com)
Myeong Soo Lee (drmslee@gmail.com)
Edzard Ernst (edzard.ernst@pms.ac.uk)

Version: 5 Date: 15 May 2012

Author's response to reviews:

Dear Reviewers

We appreciate the more comments for improving our manuscript “Acupuncture for the Treatment of Tinnitus: A Systematic Review”. According to the comments my colleagues and me made the further correction.

Yours faithfully,

Myeong Soo Lee, PhD